Author's response to reviews

Title: The effectiveness of peer health coaching in improving glycemic control among low-income patients with diabetes: protocol for a randomized controlled trial

Authors:

Amireh Ghorob (ghoroba@fcm.ucsf.edu)
Maria M Vivas (mamevi@yahoo.com)
Diana De Vore (devored@fcm.ucsf.edu)
Victoria Ngo (tori.ngo@gmail.com)
Thomas Bodenheimer (tbodenheimer@fcm.ucsf.edu)
Ellen Chen (elchen@fcm.ucsf.edu)
David H Thom (dthom@fcm.ucsf.edu)

Version: 3 Date: 14 February 2011

Author's response to reviews:

February 14, 2011

Dear Editor:

A minor revision to the article “The effectiveness of peer health coaching in improving glycemic control among low-income patients with diabetes: protocol for a randomized controlled trial” was made in the Acknowledgements section.

Sincerely,

Amireh Ghorob

Ms. Amireh Ghorob, MPH
Project Director, Peers for Progress
Master Health Coach Trainer, CEPC
UCSF Department of Family and Community Medicine
995 Potrero Avenue, Bldg. 80 WD 83
University of California, San Francisco
San Francisco, CA 94110
P 415/206-6018
F 415/206-8387
ghoroba@fcm.ucsf.edu